Author's response to reviews

Title: Understanding the modifiable health systems barriers to hypertension management in Malaysia: a multi-method health systems appraisal approach

Authors:

Isabelle Risso-Gill (isabelle.risso-gill@lshtm.ac.uk)
Dina Balabanova (dina.balabanova@lshtm.ac.uk)
Fadhlina Majid (fadhlina.majid@gmail.com)
Kien Keat Ng (dr.kng@hotmail.com)
Khalid Yusoff (k.yusoff@salam.uitm.edu.my)
Feisul Mustapha (dr.feisul@moh.gov.my)
Charlotte Kuhlbrandt (Charlotte.Kuhlbrandt@lshtm.ac.uk)
Robby Nieuwlaat (Robby.Nieuwlaat@phri.ca)
J-D Schwalm (schwalji@mcmaster.ca)
Tara McCready (tara.McCready@phri.ca)
Teo Koon (Koon.Teo@phri.ca)
Salim Yusuf (Salim.Yusuf@phri.ca)
Martin McKee (martin.mckee@lshtm.ac.uk)

Version: 3
Date: 19 June 2014

Author's response to reviews:

Health systems interventions are often applied without adequate recognition of the local context. We describe a structured method to assess health systems and its use in designing a complex intervention to improve hypertension management in a middle income country.